**Author’s response to reviews**

**Title:** Bioactivity of essential oils extracted from Cupressus macrocarpa branchlets and Corymbia citriodora leaves grown in Egypt

**Authors:**

Mohamed Salem (zidan_forest@yahoo.com)

Hosam Elansary (hosammail2003@yahoo.com)

Hayssam Ali (hayhassan@ksu.edu.sa)

Ahmed El-Settawy (a_settawy@yahoo.com)

Mohamed Elshikh (zidan_foresty@hotmail.com)

Eslam Abdel-Salam (431106719@student.ksu.edu.sa)

Krystyna Skalicka-Woźniak (krystyna.skalicka-wozniak@umlub.pl)

**Version:** 2  **Date:** 12 Nov 2017

**Author’s response to reviews:**

BCAM-D-17-00278R1

Bioactivity of essential oils extracted from Cupressus macrocarpa branchlets and Corymbia citriodora leaves grown in Egypt

Mohamed Salem, Associate Prof.; Hosam Elansary; Hayssam Ali; Ahmed El-Settawy; Mohamed Elshikh; Eslam Abdel-Salam; Krystyna Skalicka-Woźniak

BMC Complementary and Alternative Medicine

Dear Dr. Salem,

Your manuscript "Bioactivity of essential oils extracted from Cupressus macrocarpa branchlets and Corymbia citriodora leaves grown in Egypt" (BCAM-D-17-00278R1) has been assessed by our reviewers. Based on these reports, and my own assessment as Editor, I am pleased to inform you that it is potentially acceptable for publication in BMC Complementary and Alternative
Medicine, once you have carried out some essential revisions, and some minor revisions suggested by our reviewers.

The reviewer reports, together with any other comments, are below. Please also take a moment to check our website at http://bcam.edmgr.com/ for any additional comments that were saved as attachments. Please note that as BMC Complementary and Alternative Medicine has a policy of open peer review, you will be able to see the names of the reviewers.

Once you have made the necessary corrections, please submit a revised manuscript online at:

http://bcam.edmgr.com/

If you have forgotten your username or password please use the "Send Login Details" link to get your login information. For security reasons, your password will be reset.

A point-by-point response letter must accompany your revised manuscript. This letter must provide a detailed response to each reviewer/editorial point raised, describing exactly what amendments have been made to the manuscript text and where these can be viewed (e.g. Methods section, line 12, page 5). If you disagree with any comments raised, please provide a detailed rebuttal to help explain and justify your decision.

At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.

Please also ensure that your revised manuscript conforms to the journal style, which can be found at the Instructions for Authors on the journal homepage.

Please note, if your manuscript is accepted you will not be able to make any changes to the authors, or order of authors, of your manuscript once the editor has accepted your manuscript for publication. If you wish to make any changes to authorship before you resubmit your revisions, please reply to this email and ask for a 'Request for change in authorship' form which should be completed by all authors (including those to be removed) and returned to this email address. Please ensure that any changes in authorship fulfil the criteria for authorship as outlined in BioMed Central's editorial policies

(http://www.biomedcentral.com/about/editorialpolicies#authorship).
Once you have completed and returned the form, your request will be considered and you will be advised whether the requested changes will be allowed.

By resubmitting your manuscript you confirm that all author details on the revised version are correct, that all authors have agreed to authorship and order of authorship for this manuscript and that all authors have the appropriate permissions and rights to the reported data.

Please be aware that we may investigate, or ask your institute to investigate, any unauthorised attempts to change authorship or discrepancies in authorship between the submitted and revised versions of your manuscript.

A decision will be made once we have received your revised manuscript, which we expect by 10 Dec 2017.

We look forward to receiving your revised manuscript and please do not hesitate to contact us if you have any questions..

Best wishes,

Faye Nourollahi

BMC Complementary and Alternative Medicine

https://bmccomplementalternmed.biomedcentral.com/

Editor Comments:

Please make the following revisions to your manuscript:

1. Please include the email addresses of all authors on the title page of your manuscript.
   
   Answer: Done.

2. Please provide separate Results and Discussion sections, in order to adhere to journal submission requirements.
3. Thank you for providing a section on author contribution. We use this information to note which author contributed what to the study. We had noticed only two of the seven authors are mentioned in this section. Please revise this section to include all authors, so that all authors listed in the manuscript can qualify for authorship.

Answer: Done.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. We respectfully request you please amend this statement to adhere to the guidelines for authorship that are applicable in your research field or, in the absence of any guidelines, to the International Committee of Medical Journal Editors (ICMJE) guidelines. According to the ICMJE guidelines, to qualify as an author one should have:

- made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data;

- been involved in drafting the manuscript or revising it critically for important intellectual content;

- given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; and

- agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Acquisition of funding, collection of data, or general supervision of the research group, alone, does not usually justify authorship.

If you would please amend the statement to reflex these requirements, we would appreciate it.

Additionally, please use this statement to confirm that all authors have read and approved of the final manuscript.

Answer: Done

4. Please provide a separate heading for Funding, which should be in a separate paragraph from Acknowledgements.
5. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.

Answer: Done.

6. Upon uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

Answer: Done

BMC Complementary and Alternative Medicine operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

Reviewer reports:

Maria Graça Costa Miguel (Reviewer 2): It lacks a leeter (statistical) in Table 2, for Pectobacterium carotovorum

Answer: Done.

----------------------

Editorial Policies
----------------------

Please read the following information and revise your manuscript as necessary. If your manuscript does not adhere to our editorial requirements this will cause a delay whilst the issue is addressed. Failure to adhere to our policies may result in rejection of your manuscript.

In accordance with BioMed Central editorial policies and formatting guidelines, all submissions to BMC Complementary and Alternative Medicine must have a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal's Submission
Guidelines web page for information regarding the criteria for each sub-section (https://bmccomplementalternmed.biomedcentral.com/).

Where a mandatory section is not relevant to your study design or article type, for example, if your manuscript does not contain any individual persons data, please write "Not applicable" in these sections.

For the 'Availability of data and materials' section, please provide information about where the data supporting your findings can be found. We encourage authors to deposit their datasets in publicly available repositories (where available and appropriate), or to be presented within the manuscript and/or additional supporting files. Please note that identifying/confidential patient data should not be shared. Authors who do not wish to share their data must state that data will not be shared, and provide reasons for this in the manuscript text. For further guidance on how to format this section, please refer to BioMed Central's editorial policies page - http://www.biomedcentral.com/submissions/editorial-policies#availability+of+data+and+materials.

Declarations

- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information

Answer: Done